The current issue and full text archive of this journal is available on Emerald Insight at: https://www.emerald.com/insight/2044-1398.htm

CFRI 12,1

20

Received 18 June 2021 Revised 29 September 2021 Accepted 10 November 2021

Cryptocurrencies and portfolio diversification in an emerging market

Lehlohonolo Letho Department of Financial Governance, College of Accounting Sciences, University of South Africa, Pretoria, South Africa, and Grieve Chelwa and Abdul Latif Alhassan Development Finance Centre (DEFIC), University of Cape Town Graduate School of Business – Breakwater Campus, Cape Town, South Africa

Abstract

Purpose – This paper examines the effect of cryptocurrencies on the portfolio risk-adjusted returns of traditional and alternative investments within an emerging market economy.

Design/methodology/approach – The paper employs daily arithmetic returns from August 2015 to October 2018 of traditional assets (stocks, bonds, currencies), alternative assets (commodities, real estate) and cryptocurrencies. Using the mean-variance analysis, the Sharpe ratio, the conditional value-at-risk and the mean-variance spanning tests.

Findings – The paper documents evidence to support the diversification benefits of cryptocurrencies by utilising the mean-variance tests, improving the efficient frontier and the risk-adjusted returns of the emerging market economy portfolio of investments.

Practical implications – This paper firmly broadens the Modern Portfolio Theory by authenticating cryptocurrencies as assets with diversification benefits in an emerging market economy investment portfolio. **Originality/value** – As far as the authors are concerned, this paper presents the first evidence of the effect of diversification benefits of cryptocurrencies on emerging market asset portfolios constructed using traditional and alternative assets.

Keywords Alternative assets, Cryptocurrencies, Portfolio optimisation

Paper type Research paper

1. Background of the study

"Some succeeded, many failed dismally. All while the price of Bitcoin skyrocketed to \$20,000 ... then fell back down to the \$3,000 in less than a year. But what does the future hold? There are hundreds of outlandish predictions of where Bitcoin will be in the next year, five, or 10 years, but rather than making price predictions; it is interesting to hear the thoughts of an early investor" (Pollock, 2019, online). Cryptocurrency is a form of digital currency (Dandapani, 2017) that is based on cryptography. Digital currency is "electronic money that serves as an alternative currency in digital or online transactions" (Dandapani, 2017, p. 614) while cryptography is "the art of solving codes" (Hornsby, 2010, p. 355). Essentially, cryptography is the use of encryption and decryption of data and information. One can view cryptocurrency as digital currency that is built using encryption and decryption. Therefore, cryptocurrency is currency secured in its own vault using electronic puzzles and codes. Nonetheless, the concept of cryptocurrency and cryptography has been around for several years. Chaum (1983) suggested the use of cryptography within payment systems. This was to

China Finance Review International Vol. 12 No. 1, 2022 pp. 20-50 © Emerald Publishing Limited 2044-1398 DOI 10.1108/CFRI-06-2021-0123

JEL Classification — C61, G11, G32, L25

The authors are grateful for the constructive feedback from two anonymous reviewers, which improved an earlier version of the paper. All caveats apply.

protect the user of electronic payment systems from their payment data being used by third parties. He proposed an untraceable payment system using blind signature systems. The object of these untraceable payment systems was to prohibit third parties from tracking the payment details, including the time, payee and number of payments made. This was done to enable the disclosure of the payee and proof of payment only if necessary and to stop the theft of payment information while blind signatures refer to the use of passwords to encrypt and decrypt information so that the information is never compromised between the creation, transportation and delivery of the information between two parties (Chaum, 1983). This led to the creation of Digicash on 21 April 1990. Digicash was a combination of untraceable payment systems and digital currency. However, due to bad management, the entity that built Digicash went bankrupt in 1998. This resulted in the demise of Digicash (Abrar, 2014; How DigiCash Blew Everything, 1999).

Satoshi Nakamoto created Bitcoin (BTC) in 2009 (Brière *et al.*, 2015; Osterrieder and Lorenz, 2017; Taylor, 2018) that has become one of the more widely used forms of cryptocurrencies (Dandapani, 2017). As on 30 August 2018 (Figure 1), BTC formed 53.36% of the total market capitalisation for cryptocurrencies (CoinMarketCap, 2018).

As on 1 October 2018 (Table 1), BTC constituted 64.51% of the top 10 cryptocurrencies market capitalisations. Figure 1 also reveals that, on 1 August 2017, the market capitalisation of BTC decreased due to the creation of Bitcoin Cash from BTC and the entrance of other cryptocurrencies into the market. Cryptocurrencies allow for a payment system that is based on encryption rather than trust (Kim *et al.*, 2018). This means that cryptocurrency transactions are fully encrypted and cannot be breached unlike conventional financial transactions that can be hacked and breached [1].

The Modern Portfolio Theory (MPT) of Markowitz (1952, 1991) suggests that investors must ensure that they diversify their portfolio investment in securities by investing across multiple economically diverse industries. The Capital Asset Pricing Model (CAPM), efficient market hypothesis (EMH) and behavioural finance theory (BFT) are other theories that are also related to investments. The CAPM infers that an investor prefers a higher future value, rather than a lower future value of an investment (Sharpe, 1964). Moreover, the CAPM notes that there is a linear relationship between risk and return (Karceski, 2002). Fama (1970) analysed the EMH that mentions three categories of information in the market, the weak form, semi-strong form and the strong form. The BFT assesses the capital markets and the investors from a psychological and sociological perspective (Lo, 2004; Malkiel, 2015; Subrahmanyam, 2007).

The empirical studies (Brière et al., 2015; Lee et al., 2017; Osterrieder and Lorenz, 2017; Brauneis and Mestel, 2019; Trucíos et al., 2020) on cryptocurrencies have some similarities and differences in their findings. The similarities found within the studies are that BTC is very risky and that it is riskier than G10 currencies. Cryptocurrencies were also found to be more volatile than traditional and alternative investments. Other similarities found were that there was a low correlation between cryptocurrencies, including BTC and traditional and alternative investments, including G10 currencies. Cryptocurrencies, including Cryptocurrency Index (CRIX) and BTC had better returns than traditional and alternative investments, including G10 and emerging market currencies (Brière *et al.*, 2015; Lee *et al.*, 2017; Osterrieder and Lorenz, 2017; Rohrbach et al., 2017). Cryptocurrencies were found to be a good diversifier of investments (Brauneis and Mestel, 2019; Brière et al., 2015; Lee et al., 2017) and reasonable investment (Brière et al., 2015; Lee et al., 2017). The differences within the studies varied. Brière et al. (2015) discovered that Bitcoin returns differed greatly from those of hedge funds, gold and oil. The empirical studies also found that cryptocurrencies were more volatile than the annual inflation of Japan, US, Britain, South Africa and Mexico (Kim et al., 2018). Equally weighted portfolios performed better than optimal portfolios for cryptocurrency investment portfolios (Brauneis and Mestel, 2019). The Fully-Connected

Portfolio diversification in an emerging market CFRI 12,1

 $\mathbf{22}$

Percentage of Total Market Capitalization (Dominance)

Overlapping Stacked 💢 🧮



Figure 1. Percentage of total market capitalisation from April 2013 to August 2018 Source(s): Data from CoinMarketCap Global Charts (2018)

In a In a 1 Bitcoin BTC R1,619,344,831,706 64.51 17,298,562 2 Ethereum ETH R333,297,114,307 13.28 102,295,813	•
2 Ethereum ETH R333,297,114,307 13.28 102,295,813	an emerging
	market
3 XRP XRP R327,427,298,382 13.04 39,870,907,279	
4 Stellar XLM R68,876,842,252 2.74 18,789,957,655	
5 Litecoin LTC R50,263,838,354 2.00 58,530,852	23
6 Tether USDT R39,705,678,904 1.58 2,806,421,736	
7 Monero XMR R26,962,499,811 1.07 16,452,054	
8 Dash DASH R22,064,326,921 0.88 8,350,590	
9 NEM XEM R12,349,926,054 0.49 8,999,999,999	
10 Dogecoin DODG R9,879,132,427 0.39 116,363,737,525	Table 1
Source(s): Data from CoinMarketCap market capitalizations (2018). ^a Authors' own calculation, percentage of market capitalisation = (market capitalisation of the cryptocurrency divided by the sum of the total market inclusion and the 10 cmptocurrency included in $(PPX) \times 100$	10 Cryptocurrencies cluded in CRIX on 1

Neural Networks (CNN) outperformed three other portfolio management algorithms and three benchmarks and they had significantly less risk and a larger Sharpe ratio than the Passive Aggressive Mean Reversion (PAMR) (Jiang and Liang, 2017). The univariate volatility and vine copula models all estimated the value-at-risk (VaR) and Expected Shortfall (ES) well. However, the combination of the vine copula and robust GARCH models was the best estimator of VaR and ES (Truc*í*os *et al.*, 2020).

The empirical studies comparing cryptocurrencies [2] to traditional and alternative assets revealed that BTC is highly volatile and risky; it has high average returns and a very low correlation to both traditional and alternative assets; cryptocurrencies are a potentially decent investment; cryptocurrencies and emerging market currencies performed better than G10 currencies; and Bitcoin is riskier that G10 currencies (Brière *et al.*, 2015; Kim *et al.*, 2018; Lee *et al.*, 2017; Osterrieder and Lorenz, 2017; Rohrbach *et al.*, 2017). Nonetheless, very few of these empirical studies focused on emerging economies, *let al*one Africa. Some of the empirical studies included the Rand as a currency however, they did not focus on comparing the performance of African or South African listed shares to cryptocurrencies. Furthermore, these studies did not assess whether cryptocurrencies are good portfolio diversification assets, in terms of an African or South African listed shares investment portfolio. The empirical literature has not compared the performance of South African listed shares to cryptocurrencies. Herein lies a gap to study South African assets. This study seeks to address this gap in knowledge.

Against this background, this paper examines the performance of cryptocurrencies in relation to traditional and alternative investments in South Africa. Therefore, this study expands on the Modern Portfolio Theory (MPT) in terms of utilising economically diverse securities to improve the expected return in relation to the risk. It also contributes towards this theory in attesting towards using low correlating assets to diversify an investment portfolio. It underscores the expansion of the pool of knowledge and data being contributed towards assets with diversification benefits within an investment portfolio. Therefore, this research seeks to assess whether the incorporation of cryptocurrencies into the pool of available South African investments would be beneficial to investors and determine whether cryptocurrencies will improve the portfolio of traditional and alternative assets. It also seeks to assess the diversification benefits of cryptocurrencies in a portfolio of traditional and alternative investments in an emerging economy and the performance of cryptocurrencies in relation to traditional and alternative assets within an emerging economy.

CFRI 12,1

 $\mathbf{24}$

This paper focuses on the South African stock exchange for the following reasons. First, the Johannesburg Stock Exchange (JSE) is the most developed stock market in Africa (Demirgüc-Kunt and Klapper, 2012). Since the market crash of 2008, investors have sought out investments, such as alternative assets, that are not correlated to the stock market, to further diversify their portfolio investments. However, most of these alternative investments have underperformed (Srilakshmi and Karpagam, 2017). Listed equities in South Africa have also been underperforming on the JSE, which is the largest African exchange and the 19th largest exchange globally (JSE, 2019). The JSE's Top 40 index and the JSE's All Share index had a return of -5.40 and -4.93% for the 12-month period ended 5 October 2018, respectively (Trading Economics, 2018). This reflects the underperformance of listed shares in South Africa, Second, the market capitalisation of Bitcoin and over 1,000 cryptocurrencies was more than US\$400 billion in January 2018. The market capitalisation of over 1,000 cryptocurrencies excluding Bitcoin in January 2018 was more than US\$250 billion. Various cryptocurrency exchanges have daily volumes of US\$50 billion and Bitcoin futures have been created in places like Chicago so that institutional investors can hedge and trade Bitcoin. Additionally, between January 2009 and April 2017, 606 million transactions of US\$1 trillion were performed (Taylor, 2018). The question is whether cryptocurrencies can assist in alleviating the illiquidity of many African financial markets as the cryptocurrency market has had good returns in the past, even though these returns have declined in recent times. Perhaps cryptocurrencies can tackle the problem resulting from the underperformance of listed shares in South Africa. Finally, the ISE is one of two emerging stock markets in the 20 largest stock exchanges that trades in cryptocurrencies. Although South Africa is ranked as the second in the cryptocurrency adoption index countries in Africa (behind Kenya and ahead of Nigeria), it has the largest trade volumes in Peer-to-peer (P2P) cryptocurrency compared to the second emerging stock market (China) that trades cryptocurrencies (Chainalysis, 2020; Euronext, 2021; Haqqi, 2020).

The rest of the paper is organized as follows; Section 2 outlines the hypothesis development; Section 3 outlines the methodology employed in the study; Section 4 discusses the results while Section 5 presents a conclusion and policy recommendations of the findings.

2. Hypotheses development

2.1 Portfolio diversification and the modern portfolio theory

The Modern Portfolio Theory (MPT) of Markowitz (1952) suggests that investors should maximise their discounted estimated future returns and minimise their risk on securities (Fama and French, 2004; Litterman, 2003b; Malkiel, 2015; Markowitz, 1991). Markowitz (1952, 1991) further stated that, in doing so, the investors must diversify their portfolio investment in securities by investing across multiple and economically diverse industries, particularly in entities which have smaller covariances or correlations in those industries being invested in. Therefore, securities should be selected while bearing in mind how the fluctuations of other securities within the portfolio impact the chosen securities (Elton and Gruber, 1997). Furthermore, an increased diversification of the portfolio decreases the risk of the investment, but it does not eliminate all the risk (Malkiel, 2015). Within the MPT, the expected returnsvariance rule states that, the expected returns should be maximised for the variance or more than the variance, and the variance should be minimised for the expected returns or less than the expected returns (Markowitz, 1952). Consequently, the variance is the driver of the expected returns (Litterman, 2003c; Malkiel, 2015). The variance of the portfolio needs to be taken into consideration when assessing the expected returns for every investment made (Litterman, 2003c). Markowitz postulates that an optimal portfolio is one that lies on the efficient frontier (Harvey et al., 2010). The efficient frontier is the amalgamation of securities within a portfolio that maximises returns for a specific risk level of the portfolio (Mangram, 2013; Pfiffelmann *et al.*, 2016).

Portfolio diversification in an emerging market

2.2 The impact of the inclusion of different asset classes in a portfolio

The inclusion of different asset classes impacts an investment portfolio, either by improving or reducing the portfolio's performance. A study performed by Lintner (1965) on mutual funds indicated that the investors' risk after diversification is significant. It also finds that diversification and cautiously selecting assets within a portfolio cannot significantly reduce the market and residual risk. Jensen (1969) performed a study on mutual funds, where the net returns of the mutual funds in relation to the risk-return amount was below that of the market. In an examination carried out by Karceski (2002) on the cash flows of equity funds for aggressive growth, growth, growth and income, equity income and income options, it showed that there is no quantitative indication that the ownership of larger mutual funds reduces the beta risk premium, however, it showed that market returns exceeded funds flowing from stocks. Brière et al. (2015) conducted a study on Bitcoin, the Euro, Japanese Yen, emerging and developing economies' government bonds, equities, corporate and globally linked inflation bonds, listed real estate, oil, hedge funds and gold indices. The study found Bitcoin to be very volatile and risky at 176% per annum, however, it had high average returns of 404% per annum. Moreover, Bitcoin had a very low correlation to both traditional and alternative assets returns. An investigation was conducted by Lee et al. (2017) indicated that traditional investments do not perform as well as cryptocurrencies on an average daily return. The investigation also found that there was a low correlation between cryptocurrencies' returns and traditional investments' returns. Therefore, cryptocurrencies are likely to generate greater returns in relation to risk than South African traditional and alternative assets.

H1. Cryptocurrencies generate higher returns in relation to risk than South African traditional and alternative assets.

The study by Brière *et al.* (2015) also discovered that Bitcoin is a good diversifier of investments and that the inclusion of Bitcoin in an investment portfolio, at a small level, greatly increases the risk-return utility of a well-diversified investment portfolio (Brière *et al.*, 2015). Moreover, the investigation by Lee *et al.* (2017) also found that a portfolio consisting of cryptocurrencies, traditional and alternative assets outperformed a portfolio consisting of only traditional and alternative assets (Lee *et al.*, 2017). Thus, cryptocurrencies are anticipated to have diversification advantages in South African investment portfolios.

H2. Cryptocurrencies have diversification advantages in South African investment portfolios.

3. Methodology

3.1 Data and sample

This quantitative study employs traditional financial assets (stocks, bonds, currencies), alternative assets (commodities, real estate) and cryptocurrencies as the units of analysis. The data are used to assess these units of analysis. The data sourced consisted of time series data being the daily historical closing prices of financial assets and cryptocurrencies from 10 August 2015 to 31 October 2018. The data were extracted from 10 August 2015, as this was the date that the daily historical closing price of Ethereum was available from the very beginning of the trading week of traditional assets. Ethereum was only released to the

CFRI 12,1

 $\mathbf{26}$

public from 31 July 2015 (Coinmana, 2019). Referring to Table 1, Ethereum had to be included within the testing as it was the second largest cryptocurrency on 1 October 2018, and it formed 13.28% of the market capitalisation of 10 cryptocurrencies that were included in the CRIX on 1 October 2018. The data were sourced from the Iress Expert database for South African financial assets, the investing.com website for 3-Year South African government bonds, the CoinMarketCap website for cryptocurrencies and http:// crix.hu-berlin.de website for the CRIX. Iress is an Australian company that houses an electronic financial asset database and it has footprints in South Africa, Australia, Canada, New Zealand, the United Kingdom, Hong Kong and Singapore (Iress, 2019). CoinMarketCap is a well-known online cryptocurrency trading database that provides access to trading data for 289 cryptocurrencies from numerous cryptocurrency exchanges from across the globe (Coinguides, 2019; CoinMarketCap, 2018). CRIX is a German based cryptocurrency index that was constructed and is maintained by various academics. technologists and economists from Humboldt University, Singapore Management University and CoinGecko (Härdle et al., 2018b). CoinGecko is a financial trading web based database (CoinGecko, 2019). The data that were sourced was available for most periods under analysis, if not all, for most financial assets and cryptocurrencies. The following section will focus on sampling and the choice of data.

The two sets of populations that were studied were cryptocurrencies and financial assets from a South African investors' perspective. Traditional and alternative assets have been categorised under one unit of analysis, as financial assets, similar to the categorisation of the studies carried out by Brière et al. (2015) and Lee et al. (2017). The period of study was in days from 11 August 2015 to 31 October 2018. However, weekends and South African public holidays during this period were excluded, as trading of traditional assets does not occur during this period (JSE, 2018a, b, c). A non-probability sampling technique, being purposive sampling, was used. This technique allows for the selection of specific samples in order to test the constructs of the study (Gay et al., 2012; Leacock et al., 2015). The financial assets selected consisted of both traditional and alternative assets that would realistically be included in an investment portfolio. The traditional assets consisted of South African listed shares and government bonds while the alternative assets selected were gold, oil, platinum and South African listed property and resources. These alternative and traditional assets were classified in a similar manner and in line with studies by Brière et al. (2015) and Lee et al. (2017). Gay et al. (2012) posit that, with a large population, a 400 sample size would suffice, but a 500 sample size would add credence (Leacock et al., 2015). However, Lee et al. (2017) had a sample size that spanned the daily prices from 11 August 2014 to 27 March 2017, which is far greater than 500. Therefore, the sample size for this study was 800.

3.2 Description and measurement of variables

3.2.1 Returns. This paper employs the arithmetic returns, rather than geometric returns, because they balance out the downward bias with their upward bias (Levy and Sarnat, 1970). The computation of the arithmetic returns is defined as follows:

$$r_{i(t)} = \frac{P_{i(t)} - P_{i(t-1)}}{P_{i(t-1)}} \tag{1}$$

where r_{it} is the arithmetic rate of return for the period for *i*th financial asset or cryptocurrency, P_t is the price of the *i*th financial asset or cryptocurrency for period *t* and P_{t-1} is the price of the *i*th financial asset or cryptocurrency for period t - 1. These returns were calculated after the US dollar denominated financial asset and cryptocurrency prices were converted into South

African Rands. The daily arithmetic returns were computed for each of the three assets; namely cryptocurrency asset, alternative asset and traditional asset.

Brière *et al.* (2015) found that cryptocurrencies have different statistical characteristics from oil, gold and other assets. They have also been found to be more volatile but often result in greater returns than traditional assets (Brière *et al.*, 2015; Härdle *et al.*, 2018a; Srilakshmi and Karpagam, 2017).

This study seeks to determine whether cryptocurrencies will improve the portfolio of traditional and alternative assets and assess the diversification benefits of cryptocurrencies in a portfolio of traditional and alternative investments. Moreover, the study also assesses the performance of cryptocurrencies in relation to traditional and alternative assets. Similar investigations, performed by Brière *et al.* (2015), Kan and Zhou (2012) and Lee *et al.* (2017), classified independent and dependent variables, as well as benchmark and test assets based on the potential diversification benefits of either cryptocurrencies or international financial assets. The alternative, traditional and domestic assets were classified as independent variables and test assets. Therefore, this study followed the same mode of classification from a South African investor's perspective. Refer to Table 2 for more details as to how these assets are classified.

3.2.2 Estimation techniques. In line with Brière *et al.* (2015) and Lee *et al.* (2017), this study employs the mean-variance analysis and Sharpe ratio, conditional Value-at-risk analysis and Spanning test techniques in the analysis of the data described in Table 2. This section discusses each of the techniques as used in the study.

3.2.2.1 Mean-variance (MV) analysis and Sharpe ratio. The strength of the MV analysis is that it assesses the return in relation to the risk of each asset, while considering the risk and return of the other assets within the portfolio of investments. This is done to optimise the portfolio of investments (Litterman, 2003c; Malkiel, 2015). Its weaknesses are that it is restricted as it does not account for high moments risks due to skewness and kurtosis, and it results in large asset weights for unconstrained mean-variance optimal portfolios (Lee *et al.*, 2017; Litterman, 2003a). The approach used for the MV analysis was the expected utility function that was utilised to construct the mean-variance efficient frontier, which is represented as follows:

$$E[u(r_p(\omega))] = E[r_p(\omega)] - \lambda E[(r_p(\omega) - E[r_p(\omega)])^2]$$
⁽²⁾

and the MV was optimised using:

$$\max_{\omega \in \Omega} \Phi(\omega) := \omega_{p^{(\omega)}} - \lambda \sigma_p^2(\omega) = \alpha^T \omega - \lambda \omega^T \Sigma \omega$$
(3)

where ω is the weight of the funds invested in asset *i*, r_{ρ} is the total return of the portfolio, $u(r_{\rho}(\omega))$ is the utility function that measures $r_{\rho}(\omega)$, λ is the risk-aversion coefficient that

Asset types	Asset names	Variables descriptions	
Traditional assets Alternative assets Cryptocurrencies	JSE All Share, JSE Top 40, JSE Alternative Exchange and South African government bond JSE SA Listed Property, JSE SA Resources, gold, Platinum, Brent Crude oil and South African Rand CRIX, Bitcoin, Ethereum, XRP, Dash, NEM, Stellar, Litecoin, Dogecoin, Tether and Monero	Daily returns of the assets Daily returns of the assets Daily returns of the assets	Table 2 Variable description asset names and type

Portfolio diversification in an emerging market CFRI 12,1

 $\mathbf{28}$

accounts for the deviation of the returns from the mean, Φ is the risk-adjusted expected return function and σ_p^2 is the standard deviation of the portfolio (Tutuncu, 2012). During the analysis, either the Sharpe ratio was maximised or a particular risk rate was targeted (Jorion, 1992; Kan and Zhou, 2012). The MV analysis was computed using Matlab and there was no shorting of investments.

The Sharpe ratio (SR) is a performance measurement technique that is represented using the following formula:

$$SR_i = \frac{\mu_i - R_f}{\sigma_i} \tag{4}$$

where μ_i is the mean or the return of the investment or portfolio, R_f is the risk-free rate of return and σ_i is the investment or portfolio volatility (Cowell, 2013; Howard and Lax, 2003).

The Sharpe ratio determines how much excess return an investment provides above the riskfree rate for each unit of volatility. Although the Sharpe ratio is a performance measurement technique that is often used, it does have some weaknesses. The first weakness is that it only considers asset risks and returns and it does not account for liabilities. It does not differentiate between downside and upside volatility. Another weakness is that extremely high returns result in a greater increase of σ_i than of $\mu_i - R_i$. It does not perform well for returns that have nonnormal distributions (Howard and Lax, 2003; Rollinger and Hoffman, 2013).

3.2.2.2 Conditional value-at-risk (CVaR) analysis. The auxiliary function used to construct the CVaR efficient frontier is as follows:

$$F_{\alpha}(\omega,\gamma) := \gamma + \frac{1}{1-\alpha} \int_{f(\omega,r) \ge \gamma} (f(\omega,r) - \gamma) p(r) \mathrm{d}r$$
(5)

which allowed the study to set a target risk for the efficient frontier for a portfolio of investments with *n*-dimensional vector below:

$$\min_{\omega_t \in \mathbb{R}^p} CVaR_{\alpha}(\omega_t) = \min_{\omega_t \in \mathbb{R}^p, \gamma} F_{\alpha}(\omega, \gamma)$$
(6)

$$\min_{\omega_t \in \mathcal{R}^p} CVaR_\alpha(\omega_t) \tag{7}$$

$$s.t.u_{p,t}(\omega_t) = r_{\text{Target}}$$
 (8)

$$\omega_t \ 1_p = 1 \tag{9}$$

$$\omega_{i,t} \ge 0 \tag{10}$$

where *r* is the returns of the assets, p(r) is the probability density function of *r*, $\varphi(\omega, \gamma) := \int_{f(\omega,r) \ge \gamma} (f(\omega, r) - \gamma) p(r) dr$ is the cumulative distribution function of \int, α is the probability level, r_{Target} is the target risk or significance level, R^P is the constraints of the portfolio, α is the probability level, $\omega_{i,t} \ge 0$ represents no shorting of assets and $\omega_t \ 1_p = 1$ represents investing 100% of the funds in the portfolio. CVaR allows this study to account for the expected amount of loss that exceeds the VaR loss (Lee *et al.*, 2017; Tutuncu, 2012). CVaR was computed using Matlab.

Measures of risk are helpful in optimising portfolios where there is uncertainty and potential losses. The CVaR and VaR are measures of risk. However, the VaR is not stable and it is not easy to utilise when there is extreme risk and when losses are non-normal. Another weakness of VaR is that it does not provide for losses that can be made beyond the amount specified by VaR. VaR is more geared towards the optimisation of portfolios than allowing for conservative risk management of a portfolio. The CVaR overcomes the weaknesses of the VaR (Rockafellar and Uryasev, 2002). The CVaR's advantages are that it can be utilised to measure large amounts of risk, estimate the chances of the target return being fulfilled and the CVaR, which is also known as the Expected Shortfall (ES) and Expected Tail Loss (ETL), shows the expected extreme loss of an investment (Cowell, 2013). In portfolio optimisation, the CVaR allows for the minimization of the risk, while targeting a specific return. It also allows for the minimization of the return based on a constraint of the CVaR and it permits the maximization of the CVaR in relation to the return (Rockafellar and Uryasev, 2002). Its disadvantage is that it does not naturally account for the higher moments, being skewness and kurtosis (Amédée-Manesme *et al.*, 2019; Lee *et al.*, 2017). If a specific return is the targeted return for both the MV and CVaR analysis, the composition of the portfolio will be the same using both research designs. However, the weights within the portfolios will differ between the MV and CVaR analysis (Miskolczi, 2016).

3.2.2.3 Spanning test. To assess the performance of cryptocurrencies in relation to alternative and traditional assets within a portfolio of investments, the regression test below was used.

$$R_{2t} = \alpha + \beta R_{1t} + \varepsilon_t \quad t = 1, 2, \dots T \tag{11}$$

where R_{2t} is the *N*-vector returns on *N* test assets (being the respective cryptocurrencies), R_{1t} is the *K*-vector returns on the *K* benchmark assets (being the alternative and traditional assets), ε_t is the error term, α is a coefficient and *t* represents time which is in days (Brière *et al.*, 2015; Kan and Zhou, 2012).

The expected returns of N test + K benchmark assets are:

$$\boldsymbol{\mu} = \boldsymbol{E}[\boldsymbol{R}_t] = \begin{bmatrix} \boldsymbol{\mu}_1 \\ \boldsymbol{\mu}_2 \end{bmatrix} \tag{12}$$

and the covariance matrix of N test + K benchmark assets is:

$$V = \operatorname{Var}[R_t] = \begin{bmatrix} V_{11} & V_{12} \\ V_{21} & V_{22} \end{bmatrix}$$
(13)

and

$$\Sigma = V_{22} - V_{21} V_{11}^{-1} V_{12} \tag{14}$$

In obtaining the "exact distributions of the test statistics" that are assumed "conditional on R_{1t} , the disturbances ε_t are independent and identically distributed as multivariate normal with mean zero and variance Σ " (Kan and Zhou, 2012, p. 144). The regression model is based on the CAPM and MPT (Cowell, 2013). The cryptocurrencies, alternative assets and traditional assets form part of the test and benchmark assets.

The spanning test that was used is the Huberman and Kandel (1987) multivariate OLS regression Wald test, which was computed using SAS. The strength of Huberman and Kandel's (1987) multivariate OLS regression Wald test is that it is easier to perform than its counterparts, the Lagrange Multiplier test and the Likelihood Ratio test. The Lagrange Multiplier (LM) or the Likelihood Ratio (LR) test are not completely equivalent in rejecting or accepting the null hypothesis. The Lagrange Multiplier (LM) test performs better under asymptotic spanning tests but only when T is not tiny and the Wald test performs better for finite samples in testing. However, its disadvantage is that it sometimes results in a Type I error. T is the amount of time in a time series (Kan and Zhou, 2012). Therefore, the Wald test was chosen due to the statistical software capabilities of this test.

"The necessary and sufficient conditions for spanning in terms of restrictions" (Brière *et al.*, 2015, p. 368; Kan and Zhou, 2012, p. 142) are:

Portfolio diversification in an emerging market

 $\mathbf{29}$

30

$$H_0: \alpha = 0 \text{ and } \beta \mathbf{1}_k = \mathbf{1}_N \tag{15}$$

where (15) suggests the mean-variance spanning and implies that, for each *N* test asset, there is a portfolio of the *K* benchmark assets that has the same mean but smaller variance than the test assets (Brière *et al.*, 2015; Kan and Zhou, 2012). The Wald statistic test has the asymptotic distribution:

$$W = T(\lambda_1 + \lambda_2) \sim \chi_{2N}^2 \tag{16}$$

where:

$$\lambda_1 = \max_r \frac{1 + \theta_2^2(r)}{1 + \theta_1^2(r)} - 1 \tag{17}$$

$$\lambda_2 = \max_r \frac{1 + \theta_2^2(r)}{1 + \theta_1^2(r)} - 1 \tag{18}$$

and $\theta_2^2(r)$ and $\theta_1^2(r)$ are the Sharpe ratios for the risk-free rate, while the portfolio consists of *K* benchmark assets and *N* test assets (Brière *et al.*, 2015; Kan and Zhou, 2012).

N = 1 and thus the *F*-test is:

$$\left(\frac{1}{U}-1\right)\left(\frac{T-K-1}{2}\right) \sim F_{2,T-K-1}$$
 (19)

The regression test is run multiple times with N = 1. This is done by running the regression multiple times with different cryptocurrencies and CRIX, to assess whether each cryptocurrency has diversification advantages. This was done as CRIX is overwhelmed by Bitcoin, with Bitcoin making up 64.51% of CRIX (Table 1).

3.3 Research reliability and validity

There is validity confidence that other researchers performing similar tests with the same data will obtain the same results as this study. The data sourced were verified using the author's reputation, the reputation of the source, the period over which the data related, the plausibility of this time collection period and the location from which it was sourced. Additionally, the data were consistent with the data used in the studies performed by Brière *et al.* (2015) and Lee *et al.* (2017). This study is reliable as it is based on two rigorous studies performed and published by Brière *et al.* (2015) and Lee *et al.* (2015) was on BTC, traditional and alternative assets from an American perspective. While the study performed by Lee *et al.* (2017) was on CRIX, 10 of the most frequently used cryptocurrencies within CRIX, traditional and alternative assets. The aforementioned studies performed were consistent. The reliability of this study is further supported by using CRIX, 10 cryptocurrencies used within CRIX, traditional and alternative assets from a South African perspective within this study, which is consistent with the two previous rigorous studies performed.

4. Results

4.1 Trend analysis

In Figure 2, log cumulative returns are used for the trend analysis similar to studies by Brière *et al.* (2015) and Lee *et al.* (2017), these studies used weekly returns and average daily returns



Portfolio diversification in an emerging market



Figure 2. Cumulative returns of CRIX, traditional and alternative assets CFRI 12,1

32

for all the testing performed other than for the trend analysis respectively. This directed the use of log cumulative returns in the trend analysis and arithmetic rate of return based on the daily historical closing prices being used for all the other tests performed in this study. Cumulative returns are calculated over days, weeks, months or years. Similar to the study conducted by Brière *et al.* (2015) that used the same period for the trend analysis and the testing performed for all the other tests conducted in the study, this study uses same period for the trend analysis and the testing performed for all the other tests conducted in the study.

In Figure 2, the daily log cumulative returns graph of CRIX and the traditional and alternative assets from 11 August 2015 to 31 October 2018 was created to analyse the trend of the various financial assets over time. As per Figure 2, the returns of the assets oscillate and are symmetric around zero. The returns of bonds, property and ISE Alternative Exchange index (AtlX) move mainly between 0.05 and -0.05, very close to zero, below 0.05 and -0.05 and 0.02 and -0.02 respectively. These also show that the returns of bonds, property and AtlX are less volatile and lower on average than those of the other assets. The returns of JSE All Share index (ALSI), gold, platinum, oil, resources, Top 40 and ZAR move between 0.08 and -0.06 and they were more volatile and higher on average than bonds, property and AltX. The CRIX returns moved between 0.2 and -0.2 for most of the period and 0.3 and -0.3 between late 2017 and early 2018 and then moved back to between 0.2 and -0.2 after early 2018, after significant price adjustments in cryptocurrencies. Thus, CRIX has the highest volatility and returns in comparison to the other assets. As found by Kim et al. (2018), there was an upswing in the BTC and ETH price that was caused by BTC and ETH being used to buy other cryptocurrencies in 2017. As per a study by Lee et al. (2017), cryptocurrencies average daily returns exceeded those of traditional assets, while a study by Brière *et al.* (2015) found that BTC cumulative weekly returns exceed those of other assets on average. Therefore, based on current and prior studies, the cumulative log returns and volatility of CRIX and cryptocurrencies, on average, exceed those of traditional and alternative assets. Descriptive statistics will now be discussed below.

4.2 Descriptive statistics

Tables 3 and 4 show the descriptive statistics of the different asset classes. The annual mean of traditional and alternative assets is low, with most of the annual means being positive or negative single digit returns on assets whereas the annual mean of CRIX and cryptocurrencies are all positive triple digits, except for Usdt. Therefore, the annual means of CRIX and cryptocurrencies are high and far exceed those of the relatively low means of alternative and traditional assets. This suggests that the annual returns of cryptocurrencies are very high in relation to traditional and alternative assets. These results are supported by the studies carried out by Brière *et al.* (2015) and Lee *et al.* (2017). The very high annual returns of CRIX and cryptocurrencies supports the maximisation of the returns by the investor as per the MPT. The standard deviations and variances of the alternative and traditional assets are low in comparison to those of CRIX and cryptocurrencies, which are high. Therefore, the risk of cryptocurrencies largely exceeds that of alternative and traditional assets. The high risk of CRIX and cryptocurrencies increases instead of minimising the risk of the assets held by the investor. This increased risk contests the minimisation of risk on securities as postulated by the MPT. Most traditional and alternative assets have a skewness of between 0.5 and -0.5, with the exception being bonds with a skewness of -2.23. This means that most of these assets are relatively symmetric. Most cryptocurrencies have skewnesses of above 0.5, which means that the data is heavily skewed (McNeese, 2016). This positive skewness also illustrates that returns increase rapidly and drop at a slow pace. Similar results were found in the studies performed by Brière et al. (2015) and Lee et al. (2017). Kurtosis of most traditional and alternative assets is lower than three, with AltX, Prop and bonds being exceptions, with kurtosis exceeding 11.42. The lower than three kurtosis suggests that most of these assets' returns are platykurtic. Moreover, the

$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
% 0.01% % 1.27% % 0.03%	% 1.07% % 0.01% % - 397% % - 397% % - 397% % - 397% 800 78 1.0898 20 -0.0393 800 78 1.0898 20 -0.0393 800 78 1.0940 = JSE TC Mi = Brent Crud
n 0.01 ualised mean 1.41 ian 0.04	dard deviation 0.99 iance 0.01 imum -3.56 imum -3.56 imum -3.75 -0.08 wness -0.02 wress 1.04 lag autocorrelation -0.02 ervations 800 et a autocorrelation -0.02 ervations 800 et a s SA Government Bondes, Corrected of the second
	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$

	CRIX	BTC	ETH	XRP	Dash	NEM	XLM	LTC	Doge	Usdt	XMR
Mean	0.57%	0.53%	0.96%	0.93%	0.74%	1.57%	1.25%	0.60%	0.81%	0.03%	1.05%
Annualised mean	142.82%	132.34%	242.70%	233.47%	186.51%	396.24%	315.49%	150.55%	203.99%	6.84%	263.65%
Median	0.52%	0.45%	0.12%	-0.28%	0.13%	-0.14%	-0.68%	-0.14%	-0.10%	-0.05%	0.00%
SD	4.81%	4.83%	9.14%	9.92%	7.10%	14.20%	14.34%	7.84%	9.82%	1.35%	9.65%
Variance	0.23%	0.23%	0.84%	0.98%	0.50%	2.02%	2.06%	0.61%	0.97%	0.02%	0.93%
Minimum	-22.60%	-20.98%	-60.90%	-29.96%	-22.83%	-29.88%	-30.62%	-32.77%	-39.02%	-7.36%	-25.55%
Maximum	31.01%	30.49%	67.50%	114.97%	47.51%	169.63%	272.00%	71.57%	124.43%	6.60%	104.18%
Skewness	0.1642	0.5174	1.0421	4.1632	1.2775	5.2624	9.6281	3.3969	4.4427	(0.0437)	3.0169
Kurtosis	5.3541	5.7265	9.4996	33.6579	6.2524	49.1036	164.4675	26.4026	44.0225	2.7714	24.3495
1LA	(0.0005)	0.0015	0.0766	0.1708	(0.0110)	(0.0755)	0.0961	0.0755	0.0379	(0.0309)	0.0085
Ν	800	800	800	800	800	800	800	800	800	800	800
Note(s): $BTC = Bi$	tcoin, ETH =	: Ethereum, X.	LM = Steller,	LTC = Litect	in, $Doge = I$	logecoin, Usd	t = Tether an	d XMR = Mo	nero. $1LA =$	One-lag auto	correlation
Source(s): Data fro	om authors' c	omputations									

Table 4.Descriptive statisticsfor daily returns ofCRIX and cryptocurrencies

CFRI 12,1

kurtosis of CRIX and all the cryptocurrencies also exceeds three. This means that returns of these assets are leptokurtic and subject to fat-tailed or extreme cases of risk (Forsberg, 2019) while the low one-lag autocorrelation for all the assets reflects a lack of predictability of the returns. This finding is similar to that of Lee *et al.* (2017).

4.3 Correlation analysis

Table 5 that illustrates the Pearson correlation matrix will now be discussed. Most correlations between the cryptocurrencies and the alternative and traditional investments are below 0.1 and -0.1. This suggests that there are several weak positive and negative relationships and correlations of below 0.1, -0.1, 0.5 and -0.5 between cryptocurrencies and alternative and traditional assets. Similar results of low correlations between BTC and traditional and alternative assets, as well as CRIX and traditional and alternative assets, respectively were found in studies by Brière et al. (2015) and Lee et al. (2017). From 2011 to 2016 low correlations between BTC and traditional and alternative assets were also found in a study by Burniske and White (2017). This study found that BTC has on average a low correlation of -0.05 correlated to emerging market currencies. This study also revealed that gold had a low to moderate correlation to BTC. The low correlations between the cryptocurrencies and the alternative and traditional investments builds upon the MPT of assets and industries that can be used to diversify an investment portfolio. These results also show that cryptocurrencies are potentially good diversifiers of traditional and alternative assets. However, correlations are known to change over time and these results should be assessed with caution (Brière et al., 2015). Moving forward below, the MV analysis findings are discussed.

4.4 Mean-variance results

The results of the mean-variance analysis efficient frontier are presented in Figure 3. The figure indicates that cryptocurrencies generate higher returns in relation to risk than South African traditional and alternative assets, supporting H1. The only asset that lies on the efficient frontier is CRIX, while the bonds lie very close to the lower end of the efficient frontier. All the remaining assets, being: gold, oil, ALSI, Top 40, platinum, property, AltX, resources and ZAR, do not lie individually as assets on the efficient frontier. The assets that lie on the lowest point are property, which is followed by AltX. That means that AltX is the asset with the lowest return and with a respectively high level of risk. The efficient frontier is a line that consists of numerous optimised investment portfolios, which are a combination of the assets at different weights, risks and returns (Malkiel, 2015). Therefore, the efficient frontier is the point at which the investment portfolio return is maximised per unit of risk (Mangram, 2013; Pfiffelmann *et al.*, 2016; Watsham and Parramore, 1997).

In Figure 3, at the furthest point of the efficient frontier, CRIX is set to a portfolio investment of 100%. CRIX would be the only asset in the portfolio, that is, if the respective investor was willing to take the very high risk, for the high return at the top of the efficient frontier. If an investor were to hold the best investment to derive the highest return, it would have to be 100% CRIX, despite the high risk involved. This finding contests Markowitz (1952, 1991) who stated that investors should diversify their investment portfolios, as the above finding suggests 100% investment in CRIX. This study shows that CRIX results in a higher return per unit of risk than gold, oil, ALSI, Top 40, platinum, property, AltX, resources and ZAR. Similar results were derived by Lee *et al.* (2017) in their investigation, which also found that CRIX is on the top of the efficient frontier. However, their study was not from a South African investor's perspective. Their study found that the S&P 500, the Real Estate Investment Trusts (REITs), gold, oil, private equity and Goldman Sachs Commodity Index (GSCI) all lay below the efficient frontier, with oil being the asset with the smallest return for its respectively high level of risk.

Portfolio diversification in an emerging market

CFRI 12,1	sdt XMR	41 1 rent Crude
	loge Us	063 1 304*** 0.0 m, Oil = B
36	LTC I	72** 1 093** 0. 315** 0.: ≓=Platinu
	MJX	
	NEM	L 1.360 ⁹⁴⁶ 1 0.277 ⁴⁵⁶ 0 0.317 ⁴⁵⁶ 0 0.317 ⁴⁶⁶ 0 0.3196 ⁴⁶⁶ 0 0.1196 ⁴⁶⁶ 0
	Dash	1 0.218** 1 0.218** 0.0.28** 1 0.411** 0.0.38** 0.0.38** 0.0.38** 0.0.35** 0.0.35** 0.0.455** 0.00**** 0.00*** 0.00*** 0.00**** 0.00**** 0.00**** 0.00**** 0.00**** 0.00**** 0.00********
	XRP	1 0.242*** 0.5695*** 0.5695*** 0.012*** 0.012*** 0.012*** 0.234*** 0.234***
	ETH	1 0.196*** 0.423*** 0.161** 0.161** 0.312*** 0.312*** 0.330*** 0.330*** 0.340** 0.340**
	BTC	1 0.370*** 0.309*** 0.464*** 0.201*** 0.21*** 0.413*** 0.413*** 0.413***
	CRIX	1 0.711*** 0.317*** 0.317*** 0.317*** 0.401*** 0.418** 0.418** 0.418** 0.142*** 0.348** 0.348** 0.348** 0.348**
	ZAR	$\begin{array}{c} 1 \\ 0.197^{**} \\ 0.225^{**} \\ 0.117^{**} \\ 0.117^{**} \\ 0.125^{**} \\ 0.0191^{**} \\ 0.005^$
	Oil	1 0.284*** 0.082** -0.022 0.045 0.045 0.045 0.055 0.047 0.05 0.047** 0.047** 0.047** 0.047** 0.047**
	Plat	1 0.539*** 0.568*** 0.0568*** 0.058*** 0.058** 0.0578* 0.0578* 0.0578* 0.0377* 0.037* 0.037*
	Gold	1 0.730** 0.185** 0.155** 0.154** 0.154** 0.164** 0.164** 0.066** 0.071* 0.055** 0.065** 0.065** 0.065** 0.065** 0.065** 0.065**
	Bonds	1 -0.407*** -0.227*** -0.128*** -0.132*** -0.132*** -0.033 -0.038** -0.063 -0.046 -0.046 -0.046 -0.046 -0.046 -0.045 ** -0.045 ** -0.058
	Res	1 -0.021 -0.023 -0.078* 0.078* 0.078* 0.015 -0.015 0.003 -0.002 -0.003 0.003 -0.003 0.0016 0.003 0.0016 -0.003 0.0017 -0.003 0.007 -0.003 0.007 -0.002 0.007 -0.002 0.007 -0.007
	Prop	1 0.110** 0.425** 0.296** 0.0296** 0.0296** 0.0296** 0.0296 0.0077* 0.00170000000000
	AltX	$\begin{array}{c} 1 \\ 0.002 \\ -0.002 \\ -0.013 \\ 0.004 \\ -0.015 \\ -0.015 \\ -0.018 \\ -0.018 \\ -0.018 \\ -0.018 \\ -0.018 \\ -0.018 \\ -0.018 \\ -0.018 \\ -0.003 \\ -0.$
	Top40	1 0.0037 0.6651** 0.6551** 0.053** 0.031 0.078* 0.078* 0.078* 0.078* 0.008 0.008 0.008 0.0013 0.001 0.001 0.002 0.0023 0.001 0.0023 0.001 0.0023 0.0023 0.0023 0.0023 0.0023 0.0023 0.0023 0.00
Table 5	All	994*** 1044 1044 1084 1188*** 1188*** 1147** - 1147** - 1147** - 109** - 0014 - 0014 - 0014 - 0014 - 0014 - 0015 - 0015 - 007 007 007 007 007 007 007 00
Pearson correlation matrix		All Top40 (Abry Construction) Bonds (Res 0 Bonds (Cold - 0 Old - 0 Dash - 0 Old - 0 Old - 0 Dash -



Portfolio diversification in an emerging market

37

Figure 3. Efficient frontier The reason for CRIX resulting in better returns at a specified risk is due to it being an amalgamation of numerous cryptocurrencies which have high risks but also high returns. Cryptocurrencies are known to be volatile as their prices are susceptible to implosions or bubbles (Brière *et al.*, 2015; Hickey, 2017; Taylor, 2015).

JSE listed indices consist of various top performing listed entities on the JSE which is the 19th largest stock market globally that caters to various types of investors. The JSE has to ensure that the investments are regulated, efficient and that they are kept safe (JSE, 2019). Brière *et al.* (2015) found that gold, oil and hedge funds have different characteristics to Bitcoin. Therefore, it would be expected that the risk and return of traditional and alternative assets would be lower than those of CRIX and cryptocurrencies (see Table 6).

The Sharpe ratio measures the risk-adjusted return of an asset. From Table 6, the CRIX Sharpe ratio exceeds the Sharpe ratio of ALSI, Top 40, AltX, Prop, Res, gold, platinum, oil and ZAR. The CRIX Sharpe ratio is more than double the Sharpe ratio of ALSI, Top 40, AltX, Prop, Res, gold, platinum, oil and ZAR. However, the Sharpe ratio of the bonds exceeds that of CRIX. The risk-adjusted return of CRIX is better than that of ALSI, Top 40, AltX, Prop, Res, gold, platinum, oil and ZAR. While the risk-adjusted return of the bonds is better than that of the CRIX. In line with the research performed by Brière *et al.* (2015), the Sharpe ratio analysis shows that overall CRIX is a better investment than South African traditional and alternative assets which supports H1 and affirms the importance of analysing the variance (V) in relation to expected returns (E) for assets within a portfolio as stated by the MPT.

This study finds that, although CRIX lies on the highest point of the efficient frontier, the optimal point on the efficient frontier for an investor is known as the tangency portfolio, where the Sharpe ratio is maximised or where the investor gets the best return-risk ratio (Jorion, 1992; Kan and Zhou, 2012). This study also finds that overall CRIX has a better risk-adjusted return than traditional and alternative assets. As seen in Table 7, tangency portfolio weights are as follows: 88.42% for bonds, 2.46% for gold, 1.13% for oil, 5.54% for ZAR and 2.03% for CRIX. Therefore, due to the high-risk nature of cryptocurrencies, investors cannot invest 100% of their funds in cryptocurrencies but rather a lower portion, as per the tangency portfolio, in a portfolio that consists of cryptocurrencies, traditional and alternative assets. Due to CRIX being a cryptocurrency index, the testing of CRIX in a portfolio inadvertently tests a combination of cryptocurrencies in a portfolio. The weights of the aforementioned assets within an investment portfolio adds to the literature of the MPT and it further justifies the use of diversification of assets within an investment portfolio as postulated by Markowitz (1952, 1991) in the MPT. Additionally, unlike the studies performed by Brière *et al.* (2015) and

Names	Sharpe ratio
ISE All Share index (ALSI)	0.0901
ISE top 40 index (Top 40)	0.0747
ISE alternative index (AltX)	-0.5423
ISE SA property index (Prop)	-0.4507
ISE SA resources index (Res)	0.3381
SA government bonds (bonds)	2.4648
Gold	0.5148
Platinum	0.1346
Brent crude oil (oil)	0.7196
South Africa rand (ZAR)	0.3578
CRIX	1.8712
Source(s): Data from authors' computations	

38

Table 6.

CRIX, traditional and alternative assets' Sharpe ratios

CFRI

12.1

Names W	eights	Portfolio
JSE all share index (ALSI)	0.0000	in an emerging
JSE top 40 index (Top 40) (0 JSE alternative index (AltY)).0000	market
JSE SA property index (Prop)).0000	
JSE SA resources index (Res)).4231	
SA government bonds (bonds) 88	3.4173	39
Gold	2.4589	
Platinum (0.0000.	
Brent crude oil (oil)	1.1346	
South Africa rand (ZAR) 5	5.5372	Table 7
CRIX	2.0290	Optimal portfolio asset
Source(s): Data from authors' computations		weights

Lee *et al.* (2017), this study depicts and adds to the literature the weight at which cryptocurrencies should be included within an optimal investment portfolio.

Utilising the mean-variance analysis, the study tests H2 that cryptocurrencies have diversification advantages in South African investment portfolios. As per Figure 4, the efficient frontier of the optimised portfolio, consisting of traditional assets, alternative assets and CRIX is higher and larger than that consisting only of traditional and alternative assets.

The portfolio optimised using the Sharpe ratio within Figure 4 and Table 8 will now be discussed. Under the optimised portfolio, the annual mean of the portfolio without CRIX is 8.64% and with CRIX, it is 11.64%, while the standard deviation is 2.58% for the portfolio without CRIX and with CRIX, it is 2.93%. This means that the returns improved for a portfolio with CRIX rather than without, while the risk increased slightly. The Sharpe and Sortino ratios both increased from 3.21 to 3.76 and 5.37 to 6.47 respectively, for portfolios without CRIX to those with CRIX. The Sortino ratio that exceeds the Sharpe ratio shows that there are greater excess returns per unit of downside volatility in comparison to excess returns per unit of total volatility. Moreover, the results show that, by including CRIX within an optimised portfolio, utility can be slightly improved, although at a slightly increased risk level. A previous study performed by Lee et al. (2017) illustrated that a portfolio under the same risk level had higher returns with CRIX than without CRIX. However, unlike the investigation performed by Brière et al. (2015), this study reflects the optimal portfolio Sharpe and Sortino ratio, annual mean and standard deviation for an optimal investment portfolio with and without CRIX. This enhances the literature of the MPT and data relating to the investments pool of knowledge.

Referring to Table 8, volatility levels of 6 and 12% were used within the study similar to a study carried out by Brière *et al.* (2015). However, BTC was used by Brière *et al.* (2015) to represent cryptocurrencies, whereas this study used CRIX to represent cryptocurrencies. Although BTC constituted 64.51% of the top 10 cryptocurrencies market capitalisations as at 1 October 2018 (Table 1), CRIX is more representative of the cryptocurrency investment universe as it also consists of the bulk of other cryptocurrencies within the cryptocurrency investment universe. At a volatility of 6%, the annual mean of the portfolio without CRIX of 11.561% leaps to 20.727% for portfolios with CRIX. This means that the returns improved drastically for a portfolio with CRIX over one without CRIX. The Sharpe and Sortino ratios both increased from 1.82 to 3.08 and 2.83 to 5.11 respectively, for portfolios without CRIX to those with CRIX. This shows that, by including CRIX within a portfolio that targets a 6% volatility, utility is almost doubled. At double the volatility of 12%, the annual mean of the portfolio without CRIX of 14.93% jumps to 34.81% for portfolios with CRIX. This means that the returns improve drastically for a portfolio with CRIX over those with CRIX. This means that



Figure 4. Efficient frontier with and without CRIX

Sortino ratios both increased from 1.16 to 2.49 and 1.74 to 4.01 respectively, for portfolios without CRIX to those with CRIX. The Sortino ratio that exceeds the Sharpe ratio shows that there are greater excess returns per unit of downside volatility in comparison to excess returns per unit of total volatility. Moreover, this shows that, by including CRIX within a portfolio that targets 12% volatility, utility is more than doubled. A previous study by Brière *et al.* (2015) that compared portfolios with and without Bitcoin overall also found that, for moderately risk-averse investors, including Bitcoin in their portfolios would drastically increase their portfolios' performance but at a greatly increased risk level. Therefore, the current study establishes that including CRIX and cryptocurrencies within a portfolio of investment portfolio, although at moderate to significantly increased risk levels. It reinforces H2 as it suggests that CRIX and cryptocurrencies have moderate to significant diversification advantages in South African investment portfolios and investment portfolios. These findings further validate the MPT in that diversifying across multiple economically diverse assets and maximising the E in relation to V strengthens the portfolio of investments.

The CVaR analysis will now be assessed.

4.5 Conditional value-at-risk analysis results

The results of the CVaR analysis efficient frontier are presented in Figure 5. The figure further validates H2, as it indicates that cryptocurrencies have diversification advantages in South African portfolios. At a given return, the mean-variance efficient frontier has lesser risk than the CVaR efficient frontier at a significance level of 0.01.

This is mainly due to more risk being accounted for by the CVaR than the mean-variance analysis. This shows that, under the CVaR analysis, cryptocurrencies have diversification benefits. During their study, Lee *et al.* (2017) found that the CVaR efficient frontier at a significance level of 0.01, which included higher risk and moments of skewness and kurtosis, performs slightly worse than the mean-variance frontier. Additionally, the asset allocation was similar under the CVaR analysis. This study did not account for higher moments, as per the Cornish–Fisher expansion, due to certain research limitations but the results were similar to those of a previous study. This investigation found that, under the CVaR analysis, the utilities under the CVaR and MV analyses are similar. Therefore, cryptocurrencies have diversification advantages under the CVaR analysis. The similarity between the CVaR and MV analyses results reveals that there is a similar expected extreme loss of an investment (Cowell, 2013). The similarity between the CVaR and MV analyses results reveals that there is a similar expected extreme loss of an investment, when the South African investor includes and excludes cryptocurrencies in their investment portfolio. The results of the spanning test will now be discussed.

	Optin	nised	6% Vo	latility	12% Vo	olatility	
	Without CRIX	With CRIX	Without CRIX	With CRIX	Without CRIX	With CRIX	
Annual mean	8.641%	11.636%	11.561%	20.272%	14.934%	34.807%	
Annual standard deviation	2.579%	2.931%	6.000%	6.000%	12.000%	12.000%	
Sharpe ratio	3.2147	3.7563	1.8238	3.0775	1.1602	2.4904	Table 8
Sortino ratio	5.3741	6.4663	2.8320	5.1069	1.7406	4.0053	Efficient portfolio
CRIX (%)	0.000%	2.029%	0.000%	7.154%	0.000%	15.283%	performance with and
Source(s): Data from	n authors' comp	utations					without CRIX

Portfolio diversification in an emerging market



4.6 Spanning test results

Although the MV and CVaR analysis studies performed above illustrate that cryptocurrencies have diversification gains, the studies were performed using CRIX. At 64.51%, Bitcoin forms an overwhelming part of CRIX. As a result, at this stage, it is uncertain whether only CRIX, Bitcoin or the individual cryptocurrencies result in diversification benefits. This prompted the testing of whether the individual cryptocurrencies have diversification advantages in South African investment portfolios. Using the Huberman and Kandel (1987) multivariate OLS regression Wald test at a level of significance or alpha of 0.05, the results of the spanning test were derived. As per Table 9, the research shows that the p-values of CRIX, Bitcoin, XRP, Ethereum, Dash, NEM, Stellar and Monero were all below 0.05, as 0.0025, 0.0045, 0.0352, 0.0089, 0.0178, 0.0143, 0.0461 and 0.0106 respectively while, the p-values of Litecoin, Dogecoin and Tether were greater than 0.05 at 0.0826, 0.0678 and 0.9997 respectively. This means that, since the p-value of CRIX, Bitcoin, XRP, Ethereum, Dash, NEM, Stellar and Monero is less than an alpha of 0.05, we reject the H₀ or null hypothesis. Therefore, there is significant evidence to support the fact that CRIX, Bitcoin, XRP, Ethereum, Dash, NEM, Stellar and Monero have diversification advantages in South African investment portfolios. However, since the *p*-value of Litecoin, Dogecoin and Tether is greater than an alpha of 0.05, we do not reject the H_0 or null hypothesis. There is significant evidence showing that Litecoin, Dogecoin and Tether do not have diversification advantages. Therefore, seven out of ten cryptocurrencies and CRIX have diversification benefits in South African investment portfolios, strengthening H2.

An investigation of a similar nature was performed by Brière *et al.* (2015) and Lee *et al.* (2017). Brière *et al.* (2015) discovered that Bitcoin spans traditional and alternative assets. Therefore, Bitcoin was found to have diversification benefits in a portfolio. While, Lee et al. (2017) found that most cryptocurrencies and CRIX improve the performance of a portfolio, with six out of ten cryptocurrencies and CRIX that span and diversify portfolios constructed of traditional and alternative assets but not tangency portfolios. Although it utilised the MV analysis on each cryptocurrency and portfolio, another study carried out by Ketelaars (2018) illustrated that the inclusion of Bitcoin, Steller, XRP, Litecoin and Monero in a portfolio increased the efficient frontier and the Sharpe ratio. As a result, they each have diversification benefits. From the current and previous studies, we can conclude that CRIX and most cryptocurrencies span the traditional and alternative investment portfolios. We can also conclude that CRIX and most cryptocurrencies have diversification advantages in these portfolios, excluding tangency portfolios. These findings illustrate the importance of diversifying an investment portfolio across various economically diverse assets as per the

Asset	Alpha	F-test	<i>p</i> -value	
CRIX	0.0058	6.0200	0.0025	
Bitcoin	0.0054	5.4500	0.0045	
XRP	0.0093	3.3600	0.0352	
Ethereum	0.0102	4.7500	0.0089	
Dash	0.0071	4.0500	0.0178	
NEM	0.0142	4.2700	0.0143	T 11 0
Stellar	0.0121	3.0900	0.0461	Table 9.
Litecoin	0.0052	2.5000	0.0826	Spanning test for
Dogecoin	0.0082	2.7000	0.0678	cryptocurrencies effect
Tether	0.0000	_	0.9997	constructed of
Monero	0.0101	4.5700	0.0106	alternative and
Source(s): Data from	authors' computations			traditional assets

Portfolio diversification in an emerging market

43

CFRI 12.1 MPT. Furthermore, it illustrates the gravity of selecting securities while noting the fluctuations of other securities as stated by the MPT (Elton and Gruber, 1997).

5. Conclusion and recommendations

Traditional and alternative South African investments have been performing poorly and returning less than satisfactory returns to their investors. This study sought to assist these investors by investigating alternative assets into which they can invest. In doing so, this investigation sought to compare the return relative to risk derived from cryptocurrencies and South African traditional and alternative assets. This was achieved using the mean-variance analysis and the Sharpe ratio that found that CRIX has a higher return per unit of risk than traditional and alternative assets. This demonstrated that cryptocurrencies are a better investment than South African traditional and alternative assets. A comparison of the performance of South African investment portfolios with cryptocurrencies to portfolios without cryptocurrencies was also performed. This was achieved by comparing meanvariance with and without CRIX. It was found that including cryptocurrencies within the portfolio increases the financial gains of the portfolio. This also means that including cryptocurrencies in a portfolio leads to diversification benefits. The CVaR analysis was then performed and it illustrated that risk-return under CVaR and MV analyses are similar and that diversification advantages are still derived under the CVaR efficient frontier. Lastly, the spanning test which used the multivariate OLS regression Wald test was performed. This showed that, individually, not only CRIX but most cryptocurrencies tested spanned traditional and alternative assets.

There were various findings that were made during the study performed from 11 August 2015 to 31 October 2018. The first finding is that the cumulative log returns and volatility of cryptocurrencies exceeded those of South African traditional and alternative assets. The cryptocurrency return variables observed are CRIX, Bitcoin, Ethereum, Monero, Stellar, XRP, Litecoin, Tether, Dash, NEM and Dogecoin. The traditional asset returns variables observed were the JSE All Share index, JSE Top 40 index, JSE Alternative Exchange index and South African government bonds while the alternative asset return variables consisted of JSE SA Listed Property, JSE SA Resources, gold, platinum, Brent Crude oil and South African Rand. The second finding is that the mean and risk of cryptocurrencies was not only high but is considered extreme. Moreover, cryptocurrencies are heavily skewed, meaning that their risks increase quickly and decrease gradually. The returns of all the assets were found to be unpredictable. It was also found that most of the relationships were weak negative and positive relationships between cryptocurrencies, South African traditional and alternative assets.

In addition, CRIX, which was the representative asset of cryptocurrencies, was the only asset on the efficient frontier constructed of traditional and alternative assets that lay on the furthest top point of the efficient frontier. No alternative or traditional assets lay on the efficient frontier. The Sharpe ratio of CRIX was greater than all traditional and alternative investments, except for bonds. This revealed that overall CRIX has a better risk-adjusted return than traditional and alternative assets. The efficient frontier of South African investment portfolios including CRIX performed better than those that did not include CRIX. The Sharpe and Sortino ratios illustrated that investment portfolios with CRIX had greater utility, though at greater risk, than those without CRIX. It also showed that investment portfolio, although at far greater risk. The mean-variance efficient frontier has less risk than the CVaR efficient frontier at a given return. This is mainly due to more risk being accounted for by the CVaR than the mean-variance analysis. This study's spanning test illustrated that

CRIX, Bitcoin, XRP, Ethereum, Dash, NEM, Stellar and Monero all individually improved the South African traditional and alternative investment portfolios whereas Litecoin, Dogecoin and Tether did not. Therefore, seven out of the ten cryptocurrencies and CRIX improved the South African traditional and alternative portfolios.

Although there is a pool of knowledge on diversification, not much knowledge exists in the literature with regards to cryptocurrencies and their diversification benefits, particularly in relation to the diversification advantages of cryptocurrencies in relation to African investments. This study contributes to the MPT, by enhancing this theory by illustrating that expected returns in relation to risks need to be considered when investing assets in an investment portfolio by assessing the expected returns and risks of cryptocurrencies, alternative and traditional assets. Furthermore, this study contributes towards the MPT's need by an investor in assessing diversification of an investment portfolio using economically diverse assets, while maximising the expected return in relation to the risk of the portfolio by showing that cryptocurrencies improve the risk-adjusted returns of an investment portfolio. The study contributes towards the MPT as it validates the use of low correlation assets, being cryptocurrencies to diversify an investment portfolio. This study firmly expands on the MPT by further validating cryptocurrencies as assets with diversification advantages in a portfolio of investments. This study enhances the data on investment portfolios, cryptocurrencies and diversification of investments. Moreover, this study has financial institution and investment company implications as it reflects the need for the consideration of cryptocurrencies as diversification assets within investment portfolios of financial institutions and investment companies.

Subsequent to the period of assessment of 31 October 2018, there was a 35% decline in the cryptocurrency prices on average in November 2018 (Ouimet, 2018). This large decrease in the price of cryptocurrencies could have altered the results of the investigation performed within this study. As this period was beyond that of the study, only the possible implications and not the actual implications of the price decrease are discussed. The large price decline could have resulted in cryptocurrencies not being a better investment than South African traditional and alternative assets and it may have resulted in the inclusion of cryptocurrencies within an investment portfolio causing financial loss rather than financial gain. Another possible result of the large price decline could have been that the CRIX and individual cryptocurrencies do not result in diversification benefits within the South African investment portfolios. These are a few of the many possibilities that could have arisen from the 35% decline in the cryptocurrency prices in November 2018. Only future research can reveal the outcome of cryptocurrency price decreases on traditional and alternative assets within South African portfolios.

Notes

- 1. Please refer to studies Brière *et al.* (2015), Hickey (2017), Taylor (2015), Foley *et al.* (2018), Kim (2017), Kim *et al.* (2018) expansive discussion on the disadvantages of cryptocurrencies.
- 2. In addition, similar studies by Corbet *et al.* (2018), Kurka (2019), Liu (2019), Platanakis and Urquhart (2019), Demiralay and Bayrac (2021), Akyildirim *et al.* (2020), Bouri *et al.* (2020), Kumah and Odei-Mensah (2021) and Karim *et al.* (2021) have also focused on the frequency, time domains and connection between cryptocurrencies and traditional and other financial assets, transmission shocks to volatility; the diversification and risk management of cryptocurrencies within cryptocurrency markets; the diversification benefits of cryptocurrencies in developed economies and emerging economies; stock investment portfolio and diversification capabilities of cryptocurrencies over time within equity portfolios. While this paper is closely related to Demiralay and Bayrac (2021) and Akyildirim *et al.* (2020), this paper extends these studies by providing the first empirical evidence of the effect of diversification and alternative assets.

Portfolio diversification in an emerging market

CFRI References

Abrar, W. (2014),	Untraceable Electronic Cash with Digicash, Research(Gate, Online. doi: 10.13140/RG.2.
1.1900.3040		

- Akyildirim, E., Corbet, S., Lucey, B., Sensoy, A. and Yarovaya, L. (2020), "The relationship between implied volatility and cryptocurrency returns", *Finance Research Letters*, Vol. 33, 101212, doi: 10.1016/j.frl.2019.06.010.
- Amédée-Manesme, C.-O., Barthélémy, F. and Maillard, D. (2019), "Computation of the corrected Cornish–Fisher expansion using the response surface methodology: application to VaR and CVaR", Annals of Operations Research, Vol. 281 No. 1, pp. 423-453, doi: 10.1007/s10479-018-2792-4.
- Bouri, E., Lucey, B. and Roubaud, D. (2020), "Cryptocurrencies and the downside risk in equity investments", *Finance Research Letters*, Vol. 33, 101211, doi: 10.1016/j.frl.2019.06.009.
- Brauneis, A. and Mestel, R. (2019), "Cryptocurrency-portfolios in a mean-variance framework", *Finance Research Letters*, Vol. 28, pp. 259-264, doi: 10.1016/j.frl.2018.05.008.
- Brière, M., Oosterlinck, K. and Szafarz, A. (2015), "Virtual currency, tangible return: portfolio diversification with bitcoin", *Journal of Asset Management*, Vol. 16 No. 6, pp. 365-373, doi: 10.1057/jam.2015.5.
- Burniske, C. and White, A. (2017), "Bitcoin: ringing the bell for a new asset class", available at: https:// research.ark-invest.com/hubfs/1_Download_Files_ARK-Invest/White_Papers/Bitcoin-Ringing-The-Bell-For-A-New-Asset-Class.pdf.
- Chainalysis (2020), The 2020 Geography of Cryptocurrency Report, Chainalysis, available at: https://go. chainalysis.com/2020-geography-of-crypto-report.html.
- Chaum, D. (1983), "Blind signatures for untraceable payments", in Chaum, D., Rivest, R.L. and Sherman, A.T. (Eds), Advances in Cryptology: Proceedings of Crypto, Vol. 82, pp. 199-203, doi: 10.1007/978-1-4757-0602-4.
- CoinGecko (2019), "CoinGecko", available at: https://www.coingecko.com/en.
- Coinguides (2019), "Coin market cap explained understanding CoinMarketCap data/website", available at: https://coinguides.org/coin-market-cap-explained/.
- Coinmama (2019), "History of Ethereum", available at: https://www.coinmama.com/guide/history-ofethereum.
- CoinMarketCap (2018), "Cryptocurrency market capitalizations", available at: https://coinmarketcap.com/.
- Corbet, S., Meegan, A., Larkin, C., Lucey, B. and Yarovaya, L. (2018), "Exploring the dynamic relationships between cryptocurrencies and other financial assets", *Economics Letters*, Vol. 165, pp. 28-34, doi: 10.1016/j.econlet.2018.01.004.
- Cowell, F. (2013), Risk-Based Investment Management in Practice, 2nd ed., Palgrave Macmillan, New York, NY.
- Dandapani, K. (2017), "Electronic finance recent developments", Managerial Finance, Vol. 43 No. 5, pp. 614-626, doi: 10.1108/MF-02-2017-0028.
- Demiralay, S. and Bayracı, S. (2021), "Should stock investors include cryptocurrencies in their portfolios after all? Evidence from a conditional diversification benefits measure", *International Journal of Finance and Economics*, Vol. 26 No. 4, pp. 6188-6204, doi: 10.1002/ ijfe.2116.
- Demirgüç-Kunt, A. and Klapper, L. (2012), "Financial inclusion in Africa: an overview", available at: http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2084599.
- Elton, E.J. and Gruber, M.J. (1997), "Modern portfolio theory, 1950 to date", *Journal of Banking and Finance*, Vol. 21 Nos 11-12, pp. 1743-1759, doi: 10.1016/S0378-4266(97)00048-4.
- Euronext (2021), One Unique Marketplace Connecting Seven European Economies, available at: https:// www.euronext.com/en.

12.1

- Fama, E.F. (1970), "Efficient capital markets: a review of theory and empirical work", *The Journal of Finance*, Vol. 25 No. 2, pp. 383-417, available at: https://www.jstor.org/stable/2325486.
- Fama, E.F. and French, K.R. (2004), "The capital asset pricing model: theory and evidence", *Journal of Economic Perspectives*, Vol. 18 No. 3, pp. 25-46, doi: 10.1257/0895330042162430.
- Foley, S., Karlsen, J.R. and Putnins, T.J. (2018), "Sex, drugs, and bitcoin: how much illegal activity is financed through cryptocurrencies?", available at: https://www.law.ox.ac.uk/business-law-blog/ blog/2018/02/sex-drugs-and-bitcoin-how-much-illegal-activity-financed-through.
- Forsberg, L. (2019), "Jarque-Bera test of normality", available at: https://www.google.com/search? q=Jarque-Bera+Test+of+Normality+Lars+Forsberg+Uppsala+University&oq=Jarque-Bera+Test+of+Normality+Lars+Forsberg+Uppsala+University&aqs=chrome.69i57. 526j0j7&sourceid=chrome&ie=UTF-8.
- Gay, L.R., Mills, G.E. and Airasian, P. (2012), Educational Research: Competencies for Analysis and Applications, Johnston, J.W., Robb, C. and Bennett, P.D. (Eds), 10th ed., Pearson Education, New Jersey, New York.
- Härdle, W.K., Lee, D.K.C., Nasekin, S. and Petukhina, A. (2018a), "Tail event driven asset allocation: evidence from equity and mutual funds' markets", *Journal of Asset Management*, Vol. 19, doi: 10.1057/s41260-017-0060-9.
- Härdle, W.K., Trimborn, S., Kolesnikova, A., Chuen, D.L.K., Teo, E., Lee, T. and Petukhina, A. (2018b), "Crix", available at: http://thecrix.de/.
- Haqqi, T. (2020), 20 Largest Stock Exchanges in the World, Insider Monkey, available at: https://www. insidermonkey.com/blog/20-largest-stock-exchanges-in-the-world-826612/.
- Harvey, C.R., Liechty, J.C., Liechty, M.W. and Peter, M. (2010), "Portfolio selection with higher moments", *Quantitative Finance*, Vol. 10 No. 5, pp. 469-485, doi: 10.1080/14697681003756877.
- Hickey, M. (2017), "Cryptocurrency vs traditional investments", available at: http://cryptocrooks.com/ cryptocurrency-vs-traditional-investments/.
- Hornsby, A. (2010), "Cryptography", in Turnbull, J., Lea, D., Parkinson, D., Phillips, P., Francis, B., Webb, S., Bull, V. and Ashby, M. (Eds), Oxford Advanced Learner's Dictionary, 8th ed., Oxford University Press, Oxford, p. 355.
- How DigiCash Blew Everything (1999), available at: https://cryptome.org/jya/digicrash.htm.
- Howard, R. and Lax, Y. (2003), "Strategic asset allocation in the presence of uncertain liabilities", in Modern Investment Management: An Equilibrium Approach, John Wiley & Sons, New Jersey, New York, pp. 110-135.
- Huberman, G. and Kandel, S. (1987), "Mean-variance spanning", *The Journal of Finance*, Vol. 42 No. 4, pp. 873-888, available at: https://www.jstor.org/stable/2328296.
- Iress (2019), "Iress", available at: https://www.iress.com/.
- Jensen, M.C. (1969), "Risk, the pricing of capital assets, and the evaluation of investment portfolios", Journal of Business, Vol. 42 No. 2, pp. 167-247, available at: https://www.jstor.org/stable/2351902.
- Jiang, Z. and Liang, J. (2017), "Cryptocurrency portfolio management with deep reinforcement learning", 2017 Intelligent Systems Conference (IntelliSys), pp. 905-913. doi: 10.1109/IntelliSys.2017.8324237.
- Jorion, P. (1992), "Portfolio optimization in practice", *Financial Analysts Journal*, Vol. 48 No. 1, pp. 68-74, available at: https://www.jstor.org/stable/4479507.
- JSE (2018a), "JSE: Johannesburg stock exchange", available at: https://www.jse.co.za/.
- JSE (2018b), "JSE: JSE market holidays", available at: https://www.stockmarketclock.com/exchanges/ jse/market-holidays/2018.
- JSE (2018c), "JSE: JSE trading hours", available at: https://www.stockmarketclock.com/exchanges/jse/ trading-hours.
- JSE (2019), "JSE overview", available at: https://www.jse.co.za/about/history-company-overview.

Portfolio diversification in an emerging market

Karim, B.A., Abdul-Rahman, A., Hwang, J.Y.T. and Kadri, N. (2021), "Portfolio diversification benefits of cryptocurrencies and ASEAN-5 stock markets", <i>Journal of Asian Finance, Economics and Business</i> , Vol. 8 No. 6, pp. 567-577, doi: 10.13106/jafeb.2021.vol8.no6.0567.
Ketelaars, T. (2018), "Investing in the cryptocurrency market: analyzing the diversification effects of cryptocurrencies in a well-diversified portfolio", available at: https://theses.ubn.ru.nl/handle/ 123456789/6169.
Kim, P. (2017), <i>Why Traders Should Consider Traditional Investments over Cryptocurrency</i> , available at: https://www.benzinga.com/markets/forex/17/12/10867085/why-traders-should-consider-traditional-investments-over-cryptocurrency.
Kim, S., Sarin, A. and Virdi, D. (2018), "Crypto-assets unencrypted", Journal of Investment Management, pp. 1-49, Forthcoming, available at: https://papers.srn.com/sol3/papers.cfm?abstract_id=3117859.
Kumah, S.P. and Odei-Mensah, J. (2021), "Are cryptocurrencies and African stock markets integrated?", Quarterly Review of Economics and Finance, Vol. 81, pp. 330-341, doi: 10.1016/j.qref.2021.06.022.
Kurka, J. (2019), "Do cryptocurrencies and traditional asset classes influence each other?", Finance Research Letters, Vol. 31, pp. 38-46, doi: 10.1016/j.frl.2019.04.018.
Leacock, C.J., Warrican, S.J. and Rose, G.S.C. (2015), <i>Research Methods for Inexperienced Researchers:</i> <i>Guidelines for Investigating the Social World</i> , Ian Randle Publishers, Kingston.
Lee, D.K.C., Guo, L. and Wang, Y. (2017), "Cryptocurrency: a new investment opportunity?", <i>The Journal of Alternative Investments</i> , Vol. 20 No. 3, pp. 16-40, doi: 10.3905/jai.2018.20.3.016.
Levy, H. and Sarnat, M. (1970), "International diversification of investment portfolios", <i>The American Economic Review</i> , Vol. 60 No. 4, pp. 668-675, available at: https://www.jstor.org/stable/1818410.
Lintner, J. (1965), "Security prices, risk, and maximal gains from diversification", <i>Journal of Finance</i> , Vol. 20 No. 4, pp. 587-615, doi: 10.1111/j.1540-6261.1965.tb02930.x.
Litterman, B. (2003a), "Beyond equilibrium, the Black-Litterman approach", <i>Modern Investment Management: An Equilibrium Approach</i> , John Wiley & Sons, New Jersey, New York, pp. 76-88.
Litterman, B. (2003b), "The capital asset pricing model", <i>Modern Investment Management: An Equilibrium Approach</i> , John Wiley & Sons, New Jersey, New York, pp. 36-43.
Litterman, B. (2003c), "The insights of modern portfolio theory", <i>Modern Investment Management:</i> An Equilibrium Approach, John Wiley & Sons, New Jersey, New York, pp. 7-23.
Liu, W. (2019), "Portfolio diversification across cryptocurrencies", <i>Finance Research Letters</i> , Vol. 29, pp. 200-205, doi: 10.1016/j.frl.2018.07.010.
Lo, A.W. (2004), "The adaptive markets hypothesis: market efficiency from an evolutionary perspective", <i>The Journal of Portfolio Management</i> , Vol. 30 No. 5, pp. 15-29, doi: 10.3905/jpm. 2004.442611.
Malkiel, B.G. (2015), A Random Walk Down Wall Street: The Time-Tested Strategy for Successful Investing, 11th ed., W. W. Norton & Company, New York, New York.
Mangram M.F. (2013) "A simplified perspective of the Markowitz portfolio theory" Clobal Journal of

Kan, R. and Zhou, G. (2012), "Tests of mean-variance spanning", Annals of Economics and Finance, Vol. 13 No. 1, pp. 139-187, available at: http://aeconf.com/Articles/May2012/aef130105.pdf.

Karceski, J. (2002), "Returns-chasing behavior, mutual funds, and beta's death", Journal of Financial

and Quantitative Analysis, Vol. 37 No. 4, pp. 559-594, doi: 10.2307/3595012.

CFRI

12.1

- Mangram, M.E. (2013), "A simplified perspective of the Markowitz portfolio theory", *Global Journal of Business Research*, Vol. 7 No. 1, pp. 59-70, available at: http://search.ebscohost.com/login.aspx? direct=true&db=bth&AN=82211365&site=ehost-live.
- Markowitz, H. (1952), "Portfolio selection", Journal of Finance, Vol. 7 No. 1, pp. 77-91, doi: 10.2307/2975974.
- Markowitz, H.M. (1991), "Foundations of portfolio theory", Journal of Finance, Vol. 46 No. 2, pp. 469-477, doi: 10.1080/0142569032000127170.
- McNeese, B. (2016), "Are the skewness and kurtosis useful statistics?", available at: https://www. spcforexcel.com/knowledge/basic-statistics/are-skewness-and-kurtosis-useful-statistics.

Miskolczi,	Ρ.	(2016),	"Differences	between	mean-variance	and	mean-CVaR	portfolio	optimization
mo	dels	", availa	ıble at: https:/	/ideas.rep	oec.org/a/ora/jour	ml/v	1y2016i1p548	-557.html.	

- Osterrieder, J. and Lorenz, J. (2017), "A statistical risk assessment of bitcoin and its extreme tail behavior", Annals of Financial Economics, Vol. 12 No. 1, pp. 1-19, doi: 10.1142/S2010495217500038.
- Ouimet, S. (2018), "Bitcoin price ends November with worst monthly decline in 7 years", available at: https://www.coindesk.com/bitcoin-price-ends-november-with-worst-monthly-decline-in-7-years.
- Pfiffelmann, M., Roger, T. and Bourachnikova, O. (2016), "When behavioral portfolio theory meets Markowitz theory", *Economic Modelling*, Vol. 53, pp. 419-435, doi: 10.1016/j.econmod.2015. 10.041.
- Platanakis, E. and Urquhart, A. (2019), "Portfolio management with cryptocurrencies: the role of estimation risk", *Economics Letters*, Vol. 177, pp. 76-80, doi: 10.1016/j.econlet.2019.01.019.
- Pollock, D. (2019), "Are cryptocurrencies still worth investing in or is gold the answer?", available at: https://www.forbes.com/sites/darrynpollock/2019/03/22/are-cryptocurrencies-still-worthinvesting-in-or-is-gold-the-answer-daniel-ameduri-discusses/#18e488935f93.
- Rockafellar, T. and Uryasev, S. (2002), "Conditional value-at-risk for general loss distributions", *Journal of Banking and Finance*, Vol. 26 No. 7, pp. 1443-1471, doi: 10.1016/S0378-4266(02)00271-6.
- Rohrbach, J., Suremann, S. and Osterrieder, J. (2017), "Momentum and trend following trading strategies for currencies and bitcoin", available at: https://pd.zhaw.ch/publikation/upload/ 213002.pdf.
- Rollinger, T. and Hoffman, S. (2013), "Sortino ratio: a better measure of risk", *Futures Magazine*, February, pp. 40-42, available at: http://www.sunrisecapital.com/wp-content/uploads/2013/02/ Futures_Mag_Sortino_0213.pdf.
- Sharpe, W.F. (1964), "Capital asset prices: a theory of market equilibrium under conditions of risk", *The Journal of Finance*, Vol. 19 No. 3, pp. 425-442, doi: 10.1111/j.1540-6261.1964.tb02865.x.
- Srilakshmi, S. and Karpagam, T. (2017), "Cryptocurrency: a new asset in portfolio", in Ramachander, A., Bagrecha, C., Talur, S. and Kumar, M.V.R. (Eds), *Research in Digital Revolution and New India*, *National Conference Proceedings*, Adarsh Group of Institutions, Bangalore, pp. 88-93, doi: 10. 21095/ajmr/2017/v0/i0/122312.
- Subrahmanyam, A. (2007), "Behavioural finance: a review and synthesis", European Financial Management, Vol. 14 No. 1, pp. 12-29, doi: 10.1111/j.1468-036X.2007.00415.x.
- Taylor, T. (2015), "Bitcoin: a remarkable innovation and its limitations", available at: http:// conversableeconomist.blogspot.com/2015/05/bitcoin-remarkable-innovation-and-its.html.
- Taylor, T. (2018), "Bitcoin and illegal activity", available at: http://conversableeconomist.blogspot.com/ 2018/01/bitcoin-and-illegal-activity.html.
- Trading Economics (2018), "Stock market", available at: https://tradingeconomics.com/stocks.
- Trucíos, C., Tiwari, A.K. and Alqahtani, F. (2020), "Value-at-risk and expected shortfall in cryptocurrencies' portfolio: a vine copula–based approach", *Applied Economics*, Routledge, Vol. 52 No. 24, pp. 2580-2593, doi: 10.1080/00036846.2019.1693023.
- Tutuncu, R.H. (2012), "Recent advances in portfolio optimization", in Scherer, B. and Winston, K. (Eds), The Oxford Handbook of Quantitative Asset Management, Oxford University Press, Oxford, pp. 7-31.
- Watsham, T.J. and Parramore, K. (1997), *Qunatitative Methods in Finance*, 5th ed., International Thomson Business Press, Oxford.

Further reading

Letho, L. (2019), "Assessing the attractiveness of cryptocurrencies in relation to traditional investments in South Africa", MCom dissertation, Graduate School of Business, University of Cape Town, available at: https://open.uct.ac.za/handle/11427/30406.

in an emerging market

Portfolio diversification

CFRI 12,1	Platanakis, E. and Urquhart, A. (2020), "Should investors include bitcoin in their portfolios? A portfolio theory approach", <i>The British Accounting Review</i> , Vol. 52 No. 4, p. 100837, doi: 10.1016/j.bar.2019.100837.
	Smith, P. and Rosevear, C. (2017), "The digital revolution bitcoin and the future of digital assets", available at: https://www.blockchain.com/whitepaper/index.html.
50	Corresponding author Abdul Latif Alhassan can be contacted at: latif.alhassan@gsb.uct.ac.za; latif.alhassan@uct.ac.za

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: permissions@emeraldinsight.com